Author's response to reviews

Title: Access, use and perceptions regarding Internet, cell-phones and PDAs as a means for health promotion for people living with HIV in Peru

Authors:

Walter H Curioso (wcurioso@u.washington.edu)
Ann E Kurth (akurth@u.washington.edu)

Version: 4 Date: 6 September 2007

Author's response to reviews: see over
September 06, 2007.

Dr Maria Kowalczuk  
Assistant Editor  
BMC-series journals  
Biomed Central.

Dear Dr Kowalckuk

I wish I could read on the Internet: the cure for HIV was just discovered: Perceptions of people living with HIV in Peru towards information technology Curioso, Kurth.  
Manuscript ID: 1125595932126270

Thank you for the acceptance of this manuscript. Please find attached our revised manuscript and a detailed response to the formatting changes requested.

We look forward to hearing from you.

Yours sincerely,

Dr. Walter H. Curioso  
Universidad Peruana Cayetano Heredia.  
Tel: (511) 99633079.  
Email: wcurioso@u.washington.edu

Formatting Changes Requested.

We have made the following requested changes:

1) Removed the text 'Research article' from the title page.
2) Removed the open and close double quotation marks from the title.
3) Removed the authors' qualifications (e.g. PhD / MD) from the manuscript and submission system.
4) Changed: From: “Corresponding Author” to: “Corresponding author.”
5) We are not underlining any section headings.
6) Removed the table placement text.
7) Removed all the websites from the text, and cited from the main reference list only.
8) Deleted the additional file that contains the response to the reviewers.
9) For tables 3 and 4, we divided the tables into cells.
10) Table 1 was converted into normal text.
11) Reference 29 was updated.